Author's response to reviews

Title: Dietary total antioxidant capacity from different assays in relation to serum C-reactive protein among young Japanese women

Authors:

Satomi Kobayashi (satomikoba@m.u-tokyo.ac.jp)
Kentaro Murakami (kenmrkm@m.u-tokyo.ac.jp)
Satoshi Sasaki (stssasak@m.u-tokyo.ac.jp)
Kazuhiro Uenishi (uenishi@eiyo.ac.jp)
Mitsuyo Yamasaki (mitsuyo@nisikyu-u.ac.jp)
Hitomi Hayabuchi (s-hayabuchi@fwu.ac.jp)
Toshinao Goda (gouda@u-shizuoka-ken.ac.jp)
Jun Oka (okajun@tokyo-kasei.ac.jp)
Keiko Baba (kbaba@mie-chukyo-u.ac.jp)
Kazuko Ohki (ohki@swu.ac.jp)
Reiko Watanabe (reiwata@unii.ac.jp)
Yoshiko Sugiyamama (syoshiko7@gmail.com)

Version: 2 Date: 1 August 2012

Author's response to reviews: see over
Dr Nehme Gabriel

Editor-in-Chief
Nutrition Journal
Central Florida Health Alliance,
United States

1 August 2012

Dear Dr Gabriel,

Resubmission of manuscript:
6066506397276872
Dietary total antioxidant capacity in relation to serum C-reactive protein among young Japanese women

Thank you for your decision letter of 27 June, 2012. We are pleased that you are interested in our manuscript for possible publication in the Nutrition Journal. We appreciate the thoroughness with which the referees regarded our manuscript. We have addressed the comments of the referees and carefully revised the manuscript. We hope that it is now suitable for publication in your esteemed journal. We understand that final acceptance depends on satisfactory resolution of the points raised by the referees.

Thank you for your attention to our manuscript. We look forward to hearing from you again.

Sincerely,

Satoshi Sasaki, MD, PhD
Department of Social and Preventive Epidemiology,
School of Public Health, the University of Tokyo,
Hongo 7-3-1, Bunkyo-ku, Tokyo 113-0033, Japan.
Tel: +81-3-5841-7872;
Fax: +81-3-5841-7873;
E-mail: stssasak@m.u-tokyo.ac.jp